

## GOPEN ACCESS

**Citation:** Huang W, Yang Z, Zhang Y, Vogt T, Armstrong B, Yu W, et al. (2024) Tropical cyclonespecific mortality risks and the periods of concern: A multicountry time-series study. PLoS Med 21(1): e1004341. https://doi.org/10.1371/journal. pmed.1004341

Received: August 8, 2023

Accepted: January 3, 2024

Published: January 22, 2024

**Copyright:** © 2024 Huang et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All mortality data used in our study were obtained from a collaborative research network under a data sharing agreement and the authors are not permitted to directly share the third-party raw data used in the analyses. For information on data access, readers are asked to contact Dr Sharon Harrison (sharon.harrison@monash.edu) for information on each country's data providers. Annual gridded population was obtained from the Global Carbon Project (https://www.cger.nies.go. jp/gcp/population-and-gdp.html). Historical information on the temporal dynamics of cyclone events across the globe was collected from the **RESEARCH ARTICLE** 

## Tropical cyclone-specific mortality risks and the periods of concern: A multicountry timeseries study

Wenzhong Huang<sup>1</sup>, Zhengyu Yang<sup>1</sup>, Yiwen Zhang<sup>1</sup>, Thomas Vogt<sup>2</sup>, Ben Armstrong<sup>3</sup>, Wenhua Yu<sup>1</sup>, Rongbin Xu<sup>1</sup>, Pei Yu<sup>1</sup>, Yanming Liu<sup>1</sup>, Antonio Gasparrini<sup>3,4,5</sup>, Samuel Hundessa<sup>1</sup>, Eric Lavigne<sup>6,7</sup>, Tomas Molina<sup>8</sup>, Tobias Geiger<sup>9</sup>, Yue Leon Guo<sup>10,11,12</sup>, Christian Otto<sup>2</sup>, Simon Hales<sup>13</sup>, Farnaz Pourzand<sup>13</sup>, Shih-Chun Pan<sup>11</sup>, Ke Ju<sup>1</sup>, Elizabeth A. Ritchie<sup>14,15</sup>, Shanshan Li<sup>1\*</sup>, Yuming Guo<sup>1\*</sup>, MCC Collaborators<sup>11</sup>

 Climate, Air Quality Research Unit, School of Public Health and Preventive Medicine, Monash University, Melbourne, Australia, 2 Potsdam Institute for Climate Impact Research, Potsdam, Germany, 3 Department of Public Health Environments and Society, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 Centre on Climate Change & Planetary Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 5 Centre for Statistical Methodology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 6 Environmental Health Science and Research Bureau, Health Canada, Ottawa, Canada, 7 School of Epidemiology and Public Health, University of Ottawa, Ottawa, Canada, 8 Department Applied Physics, Universitat de Barcelona, Barcelona, Spain, 9 Deutscher Wetterdienst (DWD), Regional Climate Office Potsdam, Potsdam, Germany, 10 Department of Environmental and Occupational Medicine, National Taiwan University (NTU) and NTU Hospital, Taipei, Taiwan, 11 National Institute of Environmental Health Sciences, College of Public Health, National Taiwan University, Taipei, Taiwan, 13 Department of Public Health, University of Otago, Wellington, New Zealand, 14 School of Earth Atmosphere and Environment, Monash University, Melbourne, Australia, 15 Department of Civil Engineering, Monash University, Melbourne, Australia

¶ Membership of MCC Collaborators is provided in Supporting Information file S1 Text \* shanshan.li@monash.edu (SL); yuming.guo@monash.edu (YG)

## Abstract

## Background

More intense tropical cyclones (TCs) are expected in the future under a warming climate scenario, but little is known about their mortality effect pattern across countries and over decades. We aim to evaluate the TC-specific mortality risks, periods of concern (POC) and characterize the spatiotemporal pattern and exposure-response (ER) relationships on a multicountry scale.

## Methods and findings

Daily all-cause, cardiovascular, and respiratory mortality among the general population were collected from 494 locations in 18 countries or territories during 1980 to 2019. Daily TC exposures were defined when the maximum sustained windspeed associated with a TC was  $\geq$ 34 knots using a parametric wind field model at a 0.5° × 0.5° resolution. We first estimated the TC-specific mortality risks and POC using an advanced flexible statistical framework of mixed Poisson model, accounting for the population changes, natural variation,

#### IBTrACS data (https://www.ncei.noaa.gov/ products/international-best-track-archive).

Funding: WH and KJ were supported by China Scholarship Council funds (nos.202006380055 and 202006240087); ZY and WY were supported by a Monash Graduate Scholarship and a Monash International Tuition Scholarship; YZ was supported by NHMRC e-Asia Joint Research Program Grant (GNT2000581); TV and CO acknowledged support from the German Federal Ministry of Education and Research (BMBF) under the research project QUIDIC (01LP1907A); RX was supported by VicHealth Postdoctoral Research Fellowships 2022; SL was supported by an Emerging Leader Fellowship of the Australian National Health and Medical Research Council (GNT2009866); YG was supported by Career Development Fellowship (GNT1163693) and Leader Fellowship (GNT2008813) of the Australian National Health and Medical Research Council; AG was supported by the Medical Research Council UK (grant ID MR/R013349/1), the Natural Environment Research Council UK (grant ID NE/ R009384/1), and the EU's Horizon 2020 project, Exhaustion (grant ID 820655); SiH and FP are supported by the Health Research Council of New Zealand. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** YG is a member of the Editorial Board of PLOS Medicine. All other authors declare no competing interests.

Abbreviations: AIC, Akaike information criterion; CDC, Center for Disease Control and Prevention; CI, confidence interval; df, degrees of freedom; ER, exposure-response; IBTrACS, International Best Track Archive for Climate Stewardship; ICD, International Classification of Diseases; IDI, Integrated Data Infrastructure; MCC, Multi-Country Multi-City; ND-GAIN, Notre Dame Global Adaptation Index; POC, periods of concern; RR, relative risk; SA3, Statistical Area Level 3; SD, standard deviation; SIM, Sistema de Informação sobre Mortalidade; SE, standard error; TA, territorial authority; TC, tropical cyclone; USA, United States of America. seasonal and day of the week effects. Then, a mixed meta-regression model was used to pool the TC-specific mortality risks to estimate the overall and country-specific ER relationships of TC characteristics (windspeed, rainfall, and year) with mortality. Overall, 47.7 million all-cause, 15.5 million cardiovascular, and 4.9 million respiratory deaths and 382 TCs were included in our analyses. An overall average POC of around 20 days was observed for TC-related all-cause and cardiopulmonary mortality, with relatively longer POC for the United States of America, Brazil, and Taiwan (>30 days). The TC-specific relative risks (RR) varied substantially, ranging from 1.04 to 1.42, 1.07 to 1.77, and 1.12 to 1.92 among the top 100 TCs with highest RRs for all-cause, cardiovascular, and respiratory mortality, respectively. At country level, relatively higher TC-related mortality risks were observed in Guatemala, Brazil, and New Zealand for all-cause, cardiovascular, and respiratory mortality, respectively. We found an overall monotonically increasing and approximately linear ER curve of TC-related maximum sustained windspeed and cumulative rainfall with mortality, with heterogeneous patterns across countries and regions. The TC-related mortality risks were generally decreasing from 1980 to 2019, especially for the Philippines, Taiwan, and the USA, whereas potentially increasing trends in TC-related all-cause and cardiovascular mortality risks were observed for Japan.

## Conclusions

The TC mortality risks and POC varied greatly across TC events, locations, and countries. To minimize the TC-related health burdens, targeted strategies are particularly needed for different countries and regions, integrating epidemiological evidence on region-specific POC and ER curves that consider across-TC variability.

## Author summary

## Why was this study done?

- Tropical cyclones (TCs), among the most destructive and costliest climate extreme events, are expected to be more intense due to climate change.
- Despite the widely acknowledged hazards, a consistent and quantitative assessment of the mortality risks of TC across countries is lacking. Such quantitative and comparable evidence across countries is urgently required to better understand the health effects and respond to the potentially increasing hazards.
- No previous studies have characterized the periods of concern (POC), exposureresponse (ER) relationship, and temporal trends of the TC health risks, directly relevant to more precise and effective preparedness and mitigation strategies.

## What did the researchers do and find?

• Using mortality data from 494 TC-exposed locations in 18 countries or territories, we quantified the TC-specific mortality risks and POC of the 382 TC events that affected

these locations. The ER relationships and temporal trends were then characterized for each country or territory.

- TC exposure was associated with a prolonged elevated risk of all-cause, cardiovascular, and respiratory mortality, with an overall average POC of around 20 days.
- The TC mortality risks and POC varied greatly across TC events, locations, and countries.
- Overall, the mortality risks increased approximately linearly with increasing TC-related maximum sustained windspeed or cumulative rainfall.
- Most studied countries or territories witnessed a decreasing TC-related mortality risks from 1980 to 2019, especially for the Philippines, Taiwan, and the USA, while potentially increasing TC-related all-cause and cardiovascular mortality risks were observed for Japan.

#### What do these findings mean?

- TC events can exhibit significant variations in their risk patterns, and future risk assessments may need to better account for this large across-TC variability.
- Targeted and evidence-based disaster management and preparedness strategies need to be developed for different countries to more effectively mitigate the TC hazards.
- Key study limitations include potential exposure misclassification errors, residual confounding, and limited generalizability.

#### Introduction

Tropical cyclones (TCs), including hurricanes, typhoons, and tropical storms, dominate weather-related disaster damages [1] and pose a major threat to our society and health [2]. It has been estimated that TCs exposed 150 million people [3] and caused billions of US dollars in damages [4,5] annually worldwide. With continued growth in coastal populations and global warming, the impacts of TCs are expected to worsen due to the increasing exposed population and proportion of very intense TCs (e.g., the warmer surface ocean is likely fueling more powerful TCs with higher windspeed and precipitation) [6–8]. These indicate that TCs will likely remain an important public health concern. Quantifying their spatiotemporal health risks has important implications for understanding the health effects and helps develop strategies to mitigate and respond to the foreseen health burden.

Emerging evidence suggests an increased risk of adverse health outcomes, mostly all-cause hospitalizations or mortality, associated with TC exposure [9–16]. Except for the immediate physical impacts such as drowning and injuries, TCs also have been found to introduce persisting or delayed elevated mortality and morbidity risks, partially attributable to medical support disruptions, environmental contamination, and psychosocial stress [17]. These indirect and longer-term effects of TC could increase the cardiovascular and respiratory mortality and morbidity, which consist of a major and important part of the disease burden indirectly

attributable to TCs. However, previous studies on TC epidemiology largely focused on a single TC event (mostly Hurricane Katrina, Sandy) restricted to a single year or area (mostly in the United States of America [USA]) [13] and focused on all-cause mortality. For example, 8 studies assessed the excess mortality in Puerto Rico after Hurricane Maria [18–25], but varied greatly in the estimated number of excess mortality (point estimates of all-cause excess mortality ranged from 514 to 4,645) due to the different utilized designs (e.g., various timeframes) and tools (e.g., survey versus mortality registration). Therefore, the results of single-TC studies may not be comparable and generalize well given the high heterogeneity in TC and population characteristics, study period/design, infrastructure, and modeling approaches across studies.

To compensate for the limitation on generalizability, several more recent studies have included multiple TCs spanning more than a decade and estimated the average health effects of TC exposures at county level in the USA [9–11,26]. While these studies revealed fundamental features of TC epidemiology in the USA, the multi-TC average health effects do not account for the across-TC variability [27]. The TC-specific health effects can vary greatly depending on the characteristics of the TC events and the population's social structure and vulnerabilities. Additionally, very few studies have estimated the temporal trends in TC-related health risks, the exposure-response (ER) relationships, as well as the periods of concern (POC) of TCs, which were important aspects of strategic disaster management and resource allocation. For example, identifying patterns in the risk magnitude and the concerned periods after TCs with diverse characteristics across regions offers valuable evidence for efficiently allocating resources, optimizing preparedness efforts, and better understanding TC health effects. However, there is an overall knowledge gap in consistently exploring the spatiotemporal mortality risks associated with TCs across countries over a long timeframe.

To address these knowledge gaps, we aim to employ a recently proposed flexible statistical framework within the framework of a two-stage analysis based on a global dataset of multiple TCs and locations over long timeframes [24]. This advanced approach could account for the TC-specific POC and mortality risks, and has been shown higher accuracy and statistical power (i.e., stronger ability to detect small and persistent increases in mortality) compared to the Farrington model currently implemented by the US Center for Disease Control and Prevention (CDC) [24]. Specifically, we aim to consistently estimate the TC-specific POC and mortality risk across TCs, locations and countries, and characterize the spatiotemporal pattern of ER relationships of mortality risk with TC characteristics. Beyond all-cause mortality, we also included 2 other leading mortality outcomes, cardiovascular and respiratory mortality, to comprehensively capture and understand the health effects of TC, including the indirect effects that were largely unclear.

## Method

#### Data collection

Based on the most updated Multi-Country Multi-City (MCC) Collaborative Research Network database, we integrated a global dataset of 1,914 locations from 44 countries or territories. Among these 1,914 locations, 494 locations from 18 countries or territories that experienced at least 1 TC during the data collection period were included in the study (**Table A in S2 Text**). The details of the MCC dataset have been described in our previous work [28,29].

Specifically, for each location in the MCC network, daily counts of all-cause mortality were collected and non-external causes (International Classification of Diseases [ICD], 9th Revision codes 0–799 or ICD-10 codes A0–R99) mortality were alternatively collected when all-cause mortality was unavailable. Two major and distinct causes of death, cardiovascular (ICD-10 codes I00–I99) and respiratory (ICD-10 codes J00–J99) mortality were also collected for each

location. Cardiovascular and respiratory mortality, largely attributable to the indirect consequences of TCs, such as property loss, resource depletion, and disruptions in medical support, constitute a major part of the TC-associated mortality burden [17]. Except for the MCC data, we also collected data on all individual deaths (date, cause, and location of death) in Australia between 2009 and 2017 from the Australian Cause of Death Unit Record File [30], in New Zealand between 2000 and 2018 from the New Zealand Ministry of Health [31], in Brazil between 1996 and 2019 from the Brazil Mortality Information System (Sistema de Informação sobre Mortalidade, SIM) [32], and in Canada between 1986 and 2015 from the Vital Statistics Deaths Database of Statistics Canada [33]. For a time-series analysis, we aggregated these individual death data at location and daily level based on the administrative boundary with a proper area size for each country (Statistical Area Level 3 [SA3] for Australia [n = 316], territorial authority [TA] for New Zealand [n = 63], immediate region for Brazil [n = 510], and second-level administrative divisions [regions or districts within the provinces and territories] for Canada [n = 288]). MCC locations in Australia, New Zealand, Brazil, and Canada were thus excluded to avoid duplication. Consequently, the integrated global dataset covers 1,914 locations from 44 countries or territories, of which 494 locations from 18 countries or territories that experienced at least 1 TC during the data collection period were included. Among these 494 locations, the all-cause mortality data in 13 locations (2.6%) was represented by non-external mortality.

To estimate the annual population for each location, the annual gridded population per 10 years between 1980 and 2100 at a spatial resolution of  $0.5^{\circ} \times 0.5^{\circ}$  (about 55 km grid), derived by ensemble learning technique and models (R-squared values ranged from 0.81 to 0.84), was also collected from the Global Carbon Project [34]. The population data were first interpolated with a natural spline function of the available values to each year for each grid [29]. The annual population of each location was then calculated as the sum of the population of the grid cells in that year covered by that location.

#### **Exposure assessment**

We used the improved Holland wind field model [35] to estimate the global historical temporal dynamics of the windspeed associated with TCs, which has been successfully applied in previous studies [8,36,37]. The methodology of this model has been described in detail in our previous work [3]. Briefly, we first obtained historical information on TCs including the position (i.e., center latitude and longitude coordinates), surface central pressure, radius, and the maximum sustained windspeed from the International Best Track Archive for Climate Stewardship (IBTrACS), a collection of best track data of TCs from sources worldwide [38]. The above variables served as inputs for the Holland wind field model as implemented within the CLIMADA Python package, an open-source impact modeling framework available on GitHub (https://github.com/CLIMADA-project/climada\_python) [39]. We generated the daily wind profile (i.e., the grid-level daily 1-min sustained wind speeds associated with the cyclone) for each cyclone event in IBTrACS from January 1st, 1980 to December 31st, 2019, at a spatial resolution of  $0.5^{\circ} \times 0.5^{\circ}$ . The estimated global historical TC-related windspeed showed a good agreement in the validation analysis of reported wind fields in the regional dataset (Pearson correlation of r = 0.86). For each location, we defined TC exposure days as days with a TCrelated maximum sustained wind speed  $\geq$  34 knots (17.5 m/s, 63 km/h, 39 mph; gale-force wind on the Beaufort scale) for the grid cell of the location [9,10]. For each TC in each location, we defined the TC hit day,  $t_0$ , as the first day with a sustained wind speed  $\geq$  34 knots for that TC in that location. We obtained the cumulative rainfall (mm) at  $t_0$  for each location from the ERA-5 reanalysis data, which is created by assimilating historical weather data from numerous

platforms (e.g., satellite, ground-based stations, radar, boats, airplanes, buoys) using sophisticated data assimilation models [40] and has been widely used in previous studies [41–45], as an additional metric of TC exposure besides the maximum sustained windspeed.

#### Statistical analysis

We adopted a two-stage analytical approach to characterize the association between TC exposure and mortality. In the first stage, we implemented a flexible statistical framework that permitted both the surveillance of concerning increases in mortality rates and careful characterization of the effect of a past event to estimate the POC and excess mortality for each TC in each location [24,46,47]. The theoretical framework and methodological details have been well described elsewhere [24]. Briefly, the daily death counts were modeled with the following mixed Poisson model with a log-link:

$$Y_{it}|\boldsymbol{\varepsilon}_{it} \sim \text{Poisson}(\mu_{it}[1+f_i(t)]\boldsymbol{\varepsilon}_{it}) \tag{1}$$

$$\mu_{it} = \text{offset}(N_{iv})e^{\alpha_i(t) + s_i(t) + w_i(t)} \tag{2}$$

In the core formula of Eq (1),  $Y_{it}$  and  $\mu_{it}$  represent the observed and expected deaths on day t in location i, respectively; (t) refers to the relative increase in mortality on day t in location i due to a TC and  $(t)^*100$  is the percent increase; and  $\varepsilon_{it}$  is a time series of auto-correlated random variables representing natural variability on day t in location i. The expected deaths,  $\mu_{it}$ can be further decomposed according to Eq (2), where  $N_{iy}$  is the population on year y in location *i* and the log of population is treated as an offset to account for the change in population size;  $\alpha_i(t)$  is a smooth function of time that accounts for the secular changes (i.e., a slow-moving trend); and  $w_i(t)$  and  $s_i(t)$  are day-of-week effects and a yearly periodic function representing a seasonal trend, respectively. During the periods without TC exposure (i.e., the control periods), we assumed (t) = 0. When different from 0, (t) was assumed to be smooth enough to be represented by a smoothing cubic spline that provides enough flexibility to detect unusual mortality fluctuation due to an extreme event like a TC. To estimate the component of interest, (t), the  $\mu_{it}$  and correlation structure of  $\varepsilon_{it}$  were first estimated based on the mortality during the control periods. Then, the  $f_i(t)$  and the standard error (SE) were estimated using the Central Limit Theorem approximation that assumed  $f_i(t)$  followed a normal distribution and accounts for the uncertainty in the expected mortality rate. Using estimates of  $f_i(t)$  and the SE, the POC was defined as a post-TC period during which a percent increase of 0 is not in the 95% confidence interval (CI) for  $f_i(t)$  (i.e., the lower limit of the 95% CI of the estimated excess mortality is greater than 0).

We permitted a discontinuity on the TC hit day, t<sub>0</sub>, to account for a sudden direct effect and fitted a smoother spline, with 6 knots per year in the main model, to provide more power to detect subtle indirect effects [24]. If a location was exposed to multiple TCs during the data collection period, we excluded the two-month (60 days) post-TC periods to exclude the effects of other TCs. The final results for the first stage were presented as the POC and excess mortality (with 95% CI) for each TC in each location. The relative risk (RR) was also calculated as the observed deaths divided by the expected deaths (i.e., observed deaths minus excess deaths) for each TC. Sensitivity analyses by excluding a different length of post-TC period (30 days, 90 days) were conducted to test the robustness of the results.

In the second stage, with the TC-specific RRs (with 95% CI) from the first stage, we further used a mixed meta-regression model, accounting for the hierarchical structure of the RRs (a location could have several TC-specific RRs for multiple TCs, i.e., TCs nested within

locations), to characterize the ER relationships of TC-related mortality risks with TC characteristics. Specifically, we built a univariate meta-regression model between the TC-specific RRs and the TC-related maximum sustained wind, cumulative rainfall, and the calendar year of TC hit, respectively, to examine the univariate ER relationship. All of these terms were included in the meta-regression models with a natural spline function of 2 degrees of freedom (df), as determined by a minimum Akaike information criterion (AIC), to allow for nonlinear ER relationships.

All data organization and analyses were conducted in R software, version 4.0.3 (Foundation for Statistical Computing, Vienna, Austria) [48]. The first- and second-stage analysis were conducted using the "excess\_model" function from the "excessmort" package [49] and the "mixmeta" function from the "mixmeta" package [50], respectively.

### Results

#### Deaths, TC, and periods of concern

The spatial distribution of the included 494 locations and a summary of these locations (e.g., study periods, number of deaths, and TCs) are shown in **Fig 1** and **Table A in S2 Text**, respectively. In total, 47.7 million all-cause deaths, 15.5 million cardiovascular deaths, and 4.9 million respiratory deaths were included in the analyses. Each location contributed an average of 21 years (standard deviation [SD]: 9.4) of data. A total of 382 TC events that hit these 494 locations during the study period were included, with an average of 7 TC events for each location (**Fig 1**). The number of exposed TCs per decade varied substantially by location, ranging from 1 to 55. Locations in Taiwan (e.g., Taipei, Kaohsiung), Japan (e.g., Naha, Okinawa), the Philippines (e.g., Manila, Valenzuela) experienced TC most frequently (average number of TC per



Fig 1. The spatial distribution and number of exposed TCs of the 494 study locations. The base layer of the world map was imported from the public domain Natural Earth project (source: https://www.naturalearthdata.com/downloads/; terms of use: www.naturalearthdata.com/about/terms-of-use/). TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g001

decade  $\geq$ 14), whereas the lowest number (1–2) were mainly observed in locations from Brazil, New Zealand, and Canada. Overall, the average length of POC after a TC for all-cause mortality was 22 days (SD: 51.3), with great variations across and within countries (**Table A in S2 Text**). The average POC were relatively longer for the study locations in the USA, Brazil, and Taiwan (>30 days), while shorter for the study locations in Vietnam, Mexico, Australia, and New Zealand ( $\leq$ 10 days). Similar overall average POC and great variations across and within countries were also observed for TC-related cardiovascular and respiratory mortality (**Table A in S2 Text**). The estimated overall and country-specific POC were robust to sensitivity analyses by excluding a different length of post-TC period (30, 60, and 90 days) (**Table B in S2 Text**).

#### TC-related excess mortality and risk

The TC-location-specific excess mortality for the 100 TCs with the highest excess deaths is shown in Fig 2, with each point indicating a location and each tick on the x-axis representing a TC (a TC with a unique ID recorded in the IBTrACS). Large variations are observed for the TC-related excess deaths within and across TCs. For example, the TC of "1999253N17124" had the highest TC-related excess deaths, which hit 1 location (Hong Kong) in China Mainland in 1999 with a maximum sustained windspeed of 40 to 45 knots and caused around 1,076 deaths (Fig 2). Despite that the TC events that contributed most respiratory and cardiovascular deaths were different from those contributed most all-cause deaths, similar patterns of large inter- and intra-TC variability were also observed (Fig 2). The TC-specific RRs varied substantially, with a point estimate ranging from 1.04 to 1.42, 1.07 to 1.77, and 1.12 to 1.92 among the 100 TCs with highest RRs for all-cause, cardiovascular, and respiratory mortality, respectively (Fig 3). A maximum RR of 1.42 (95% CI [1.09, 1.86], *p* = 0.009), 1.77 (95% CI [1.76, 1.78], p < 0.001), and 1.92 (95% CI [1.07, 3.44], p = 0.028) for all-cause, cardiovascular, and respiratory mortality were observed for TC "2011023S16147" (Australia, 2011), "2014209N16134" (Japan and South Korea, 2014), and "2011020S13182" (New Zealand, 2011), respectively. At country level, relatively higher and statistically significant TC-related mortality risks were observed in Guatemala for all-cause mortality, Brazil, Vietnam, and South Korea for cardiovascular mortality, and New Zealand and Australia for respiratory mortality (Fig 4). The countryspecific RR was generally robust to sensitivity analyses by excluding a different length of post-TC period (Table C in S2 Text).

#### ER curve of TC-related windspeed with mortality by country or territories

When characterizing the associations of TC-related maximum sustained windspeed with mortality risk, we observed an overall monotonically increasing and non-threshold curve with approximately linear shape for the ER relationships for all-cause, cardiovascular, and respiratory mortality (**Fig 5**). At country level, we found generally significant, positive linear or supra-linear ER curves of TC-related maximum sustained windspeed with mortality in Japan, South Korea, Taiwan, and the USA for all-cause mortality; Japan, Taiwan, and the USA for cardiovascular mortality; and Japan, Taiwan, and the USA for respiratory mortality (**Fig 5**). The positive ER curves were consistently observed for Japan, Taiwan, and the USA. However, there is insufficient evidence to support significant and positive ER curves between TC-related maximum sustained windspeed and mortality in other countries or regions including China Mainland, Mexico, and Thailand (**Fig 5**). Sensitivity analysis by excluding a different length of post-TC period showed robust overall and country-specific ER relationships of TC-related maximum sustained windspeed with mortality, with a similar shape to that in the main model (**Fig A in S2 Text**).



Fig 2. The top 100 TCs with highest excess deaths from all-cause, CVDs, and RDs. Each point in the figure indicates a location, and each tick on the X-axis represents a TC, which is identified by its IBTrACS event ID. A boxplot was fitted for the location-specific TC-related excess deaths within each TC. Each box represents the IQR of the excess deaths of each TC, with the middle bolded black line in the box representing the median value. The whiskers extending from the box indicate a range of 1.5 times the IQR. CVD, cardiovascular disease; IQR, interquartile range; RD, respiratory disease; TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g002

# ER curve of TC-related precipitation with mortality by country or territories

We observed similar patterns for the overall ER relationships of TC-related cumulative rainfall with mortality, with consistently significant and positive linear curves for all-cause, cardiovascular, and respiratory mortality (**Fig B in S2 Text**). Positive linear or supra-linear and monotonically increasing ER curves were detected in Japan, the Philippines, South Korea, Taiwan, and the USA for all-cause mortality; the Philippines, Taiwan, and the USA for cardiovascular



Fig 3. The top 100 TCs with highest RR for all-cause, CVDs, and RDs mortality. The RRs were estimated by comparing the deaths on TC-exposed days with those on non-exposed days, after adjusting for population changes, natural variation, seasonal, and day of the week effects. CVD, cardiovascular disease; RD, respiratory disease; RR, relative risk; TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g003

mortality; and Taiwan and the USA for respiratory mortality. Consistent monotonically increasing ER curves were consistently observed for Taiwan and the USA. No sufficient evidence of a significant and positive ER curve between TC-related cumulative rainfall and mortality in other countries or regions including China Mainland, Vietnam, Mexico, and Thailand. The estimated overall and country-specific ER relationships were robust to sensitivity analyses by excluding a different length of post-TC period (**Fig C in S2 Text**).

#### Temporal trends of TC-related mortality by country or territories

When considering the temporal variations of the TC-related mortality, an overall decreasing trend was found for all-cause and respiratory mortality, while not for cardiovascular mortality (**Fig 6**). At country level, an overall slightly increasing trend in TC-related all-cause mortality

Country or territory	All-cause mortality	RR (95% CI)	p-value	CVD mortality	RR (95% CI)	p-value	RD mortality	RR (95% CI)	p−value
USA		1.028 (1.023, 1.032)	< 0.001		1.043 (1.035, 1.052)	< 0.001	-	1.122 (1.106, 1.138)	< 0.001
Canada	101	1.044 (1.026, 1.062)	<0.001	H+H	1.055 (1.023, 1.088)	0.001		1.184 (1.130, 1.241)	<0.001
Mexico		1.034 (0.981, 1.089)	0.211		1.048 (1.002, 1.097)	0.041	F	1.095 (0.957, 1.253)	0.188
French Caribbean		1.001 (0.936, 1.070)	0.981		NA	NA		NA	NA
Brazil		1.083 (1.033, 1.135)	0.001		1.154 (1.071, 1.243)	< 0.001		1.214 (1.058, 1.392)	0.006
Philippines	H	1.018 (1.007, 1.029)	0.001	let .	1.039 (1.024, 1.055)	< 0.001	He-I	1.070 (1.025, 1.117)	0.002
China Mainland	н	0.993 (0.969, 1.017)	0.571	HHH	0.998 (0.955, 1.043)	0.938	H	0.991 (0.919, 1.068)	0.806
Taiwan		1.014 (1.005, 1.024)	0.004		1.021 (1.003, 1.040)	0.022	<b>101</b>	1.029 (0.995, 1.064)	0.100
South Korea		1.040 (1.028, 1.052)	<0.001	HeH	1.123 (1.087, 1.161)	< 0.001	H#H	1.208 (1.160, 1.258)	<0.001
Japan	•	1.011 (1.008, 1.014)	< 0.001		1.021 (1.014, 1.028)	< 0.001	-	1.051 (1.039, 1.064)	< 0.001
Thailand		0.963 (0.865, 1.072)	0.495		1.142 (0.872, 1.497)	0.335	I	1.225 (1.036, 1.447)	0.018
Vietnam	<b>↓</b>	1.146 (0.990, 1.327)	0.068	I	1.139 (1.022, 1.269)	0.019		NA	NA
Australia		1.011 (0.942, 1.085)	0.758	<b></b>	1.088 (0.994, 1.190)	0.069		1.311 (1.099, 1.563)	0.003
New Zealand	H	1.083 (1.016, 1.154)	0.014		1.106 (0.999, 1.225)	0.053		1.321 (1.142, 1.527)	< 0.001
Guatemala	H	1.145 (1.024, 1.281)	0.018		NA	NA		NA	NA
Puerto Rico	II	1.015 (0.940, 1.096)	0.697		NA	NA		NA	NA
Portugal		1.038 (0.939, 1.147)	0.467		1.331 (0.864, 2.052)	0.195		NA	NA
Spain		0.940 (0.836, 1.058)	0.307		0.781 (0.606, 1.005)	0.055	I	1.122 (0.783, 1.606)	0.530
	0.8 1 1.2 1.4	1.6		0.7 1 1.2 1.4 1	1 .6		0.7 1 1.2 1.4	ר 1.7	

Fig 4. Country or territory-specific overall RR with 95% CI for all-cause, CVDs, and RDs mortality associated with TC exposure. The RRs indicated the mortality risks in TC days compared to non-TC days. CI, confidence interval; CVD, cardiovascular disease; RD, respiratory disease; RR, relative risk; TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g004

risk was observed in Japan and Taiwan, while decreasing trends in the Philippines, South Korea, and the USA; an increasing trend in TC-related cardiovascular mortality risk was observed in Japan and Mexico, while overall decreasing trends were observed in the Philippines, South Korea, and the USA; a decreasing increasing trend in TC-related respiratory mortality risk was observed in Japan, New Zealand, the Philippines, and the USA, while a potentially increasing trend was found for Taiwan (**Fig 6**). For the remaining countries including Australia, Canada and Vietnam, Thailand, Mexico and New Zealand, no sufficient



Fig 5. The exposure-response relationship of the RR for all-cause, CVDs, and RDs mortality with TC-related maximum sustained windspeed (knots) by countries or territories. The RRs indicated the mortality risks in TC days compared to non-TC days. CVD, cardiovascular disease; RD, respiratory disease; RR, relative risk; TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g005



Fig 6. The temporal trends of the RR for all-cause, CVDs, and RDs mortality by countries or territories from 1980 to 2019. The RRs indicated the mortality risks in TC days compared to non-TC days. CVD, cardiovascular disease; RD, respiratory disease; RR, relative risk; TC, tropical cyclone.

https://doi.org/10.1371/journal.pmed.1004341.g006

evidence to detect a temporal trend. Generally, similar overall and country-specific temporal trends were observed in sensitivity analyses by excluding a different length of post-TC period (**Fig D in S2 Text**).

#### Discussion

Our large-scale population-based study estimated the TC-specific POC and mortality risks, with substantial variations in TC-related mortality risk within and across TCs. Additionally, we characterized the ER relationships and found an overall monotonically increasing, non-threshold and approximately linear curve of TC-related maximum sustained windspeed and cumulative rainfall with all-cause and cardiopulmonary mortality risks. An overall decreasing trend was observed for TC-related all-cause and respiratory mortality risk from 1980 to 2019. Further heterogeneous patterns of the ER relationships and temporal trends were revealed at the country level, such as the increasing trend in TC-related all-cause mortality risk in Japan, yet a decreasing trend in the Philippines, South Korea, and the USA.

As expected, we found that the TC-related mortality risks varied considerably across and within TCs, evidencing the necessity to account for the across-TC variability when assessing the health effects of TCs. However, among the limited epidemiological studies that systematically assessed the health effects of multiple TC events, most estimated the multi-TC average effects [9-11,26,27]. These studies also utilized a wind field model to quantify the TC exposure for a large number of TCs and found an elevated risk of mortality [9,27], hospitalization [10,11], and preterm birth [26] associated with TCs in the USA for the past decades. To our knowledge, only 1 study has estimated the TC-specific health risks, which included all TCs in the USA from 1999 to 2015 and also observed large variations in the TC-related excess mortality across and within TCs [27]. However, this study did not estimate the TC-specific POC, but instead used an 11-day post-TC period as a hypothetical POC and calculated the excess deaths within this period for all TCs. TCs could impact public health through both direct and indirect pathways. Direct impacts including physical trauma (e.g., injury and drowning) during exposure could be more immediate, while indirect impacts such as the socio-psycho environmental stress, poor or mal-nutrition due to TCs (e.g., loss of property and resources, evacuation, interruption of medical and social support) could manifest at a longer-term to increase the mortality and morbidity. Therefore, a POC of several days may not be able to sufficiently capture the

health impacts of TCs, especially for the indirect impacts, as indicated by a POC of more than 1 month for some countries/TCs in our results. Additionally, TCs could vary substantially in terms of their POC due to their different physical characteristics (e.g., windspeed, rainfall, and duration), as well as the population vulnerability. Such variability was reflected in our findings of a large standard deviation of the estimated POC across TCs, even within the same country. The self-specified identical exposure window (i.e., hypothetical POC) commonly used in previous multi-TC studies may introduce potential exposure misclassification and not be able to well capture the TC-related health effects.

To our knowledge, this study is the first to examine the ER relationships between TC characteristics and risk of mortality on a multicountry and multi-TC scale. Based on the estimated mortality risk and characteristics for each TC, we characterized the overall and country-specific ER curves of TC-related maximum sustained windspeed and cumulative rainfall with different mortality outcomes. We observed an overall monotonically increasing, non-threshold and approximately linear curve of TC-related maximum sustained windspeed and cumulative rainfall with all-cause, cardiovascular, and respiratory mortality risks. Only 1 prior study has estimated the ER curves of TC characteristics with health outcomes and observed a similar monotonically increasing ER relationship with an approximately linear shape between TCrelated maximum sustained windspeed and all-cause mortality, hospitalizations for respiratory diseases, chronic obstructive pulmonary diseases, and cardiovascular diseases in the USA [27]. Stronger TCs with higher windspeed and rainfall were more likely to induce adverse events such as flood, displacement and power outage, there having higher risks causing mortality and morbidity. Prior epidemiological studies on multiple TCs generally assessed the potential hazard of TCs only based on a binary variable of exposure (exposed versus unexposed) in terms of the maximum sustained windspeed [9-11,26,27,41,51]. The established ER curves with TCrelated windspeed and rainfall in the current study could inform the potential risks of various mortality outcomes associated with different TC intensities. Considering that more intense TCs are expected in the future under a changing climate, it is critical to incorporate the epidemiological evidence such as the ER curves in the early warning system to accurately evaluate the potential hazards of a landfalling TC and develop strategies accordingly to minimize the health burdens [52].

Heterogeneous TC-related mortality risks and patterns of the ER relationships across countries and mortality outcomes were further revealed in our study. Populations in countries like Guatemala, Brazil, Vietnam, and South Korea appear to be especially vulnerable to TC-related elevated mortality risks compared to those in other countries or regions such as Canada, Mexico, and Australia. Many factors, such as topography, economics, disaster management practices, and population characteristics, can impact the susceptibility to natural disasters like TCs [53-55]. However, there is very little evidence on systematic assessment of the vulnerability to TCs across countries and no clear explanation for such differences has yet been proposed. The higher susceptibility to the elevated mortality risks of TCs in these regions could be partially attributed to the higher frequency of high-amplitude TCs, while the relatively fewer strong TCs or study locations in other countries including Mainland China and Mexico hindered us from detecting a significant and positive ER curve [56–59]. Additionally, Clark and colleagues proposed a Notre Dame Global Adaptation Index (ND-GAIN) and attributed the differences in the overall vulnerability to climate change across countries to 6 country-level life-supporting sectors—food, water, health, ecosystem service, human habitat, and infrastructure [60]. Countries with more reliable water and food supply (e.g., higher fresh water withdrawal rate), better health and ecosystem services (e.g., less slum population and dependency on natural capital), and improved infrastructure and human habitat (e.g., less population living under 5 m above sea level and smaller age dependency ratio) could be more resilient to the

adverse impacts of climate change. The heterogeneous patterns of the ER relationships could also be explained by differences in these factors across countries. Overall, current epidemiological evidence on the potential contributing factors or effect modifiers on the associations between TCs and health is still very limited and inconclusive. More studies are required to better elucidate this issue and the underlying contributors.

We observed an overall decreasing trend for TC-related all-cause and respiratory mortality risks, with great heterogeneity across countries or territories. Despite the temporal change of the TC characteristics (e.g., tracks, frequency, intensity, duration) has been well characterized [56–59,61,62], no studies have yet estimated the temporal trend of TC-related mortality risks. To our knowledge, only 1 study examined the temporal change in the risk of homelessness, casualties, and property losses induced by TCs in South Korea and found a decreasing trend over 1979 to 2010, which is similar to our observed ER curve for TC-related all-cause mortality in South Korea [63]. Additionally, the mostly decreasing TC-related mortality risk across countries highlights the effectiveness and progress of the disaster management measures and devoted prevention efforts, especially for the Philippines, Taiwan, and the USA. While the intensity and duration of landfall TCs have been increasing [64,65], the improved early warning system and disaster preparedness practices can significantly reduce the related health risks [66–68]. However, it should also be noted that a potentially increasing trend in TC-related allcause and cardiovascular mortality risk was observed in Japan. This may be partially attributed to the considerably increasing proportion of the elderly population and the prevalence of cardiovascular diseases over the past decade in this country [69,70]. Further studies are highly warranted to elucidate the underlying mechanisms and formulate targeted approaches to reverse the increasing trends.

This study had 4 main strengths. To the best of our knowledge, this is the first and largest global investigation of the mortality risk attributed to TCs. Compared with most previous studies confined to single or several TC events within a limited region or timeframe, we collected representative death data from countries or territories of the USA (including the locations from the territories in the Caribbean [i.e., Virgin Islands and Puerto Rico]), Japan, South Korea, Canada, Brazil, Taiwan, Australia, and New Zealand. We also developed TC exposure data based on a collection of representative and best track data of TCs from official sources worldwide (i.e., IBTrACS), which has been widely used to analyze TC ecology and subsequent events (e.g., flood) [62,71-73]. A final of 382 TC events in 494 locations from 18 countries or territories during 1980 to 2019, which were characterized by different climates, socioeconomics, demographics, public health service development, and TC features. This allowed us to characterize the spatiotemporal pattern of the TC-related mortality risks, and to reduce potential selection-related biases and ensure the high-quality and generalizability of our findings. Compared to the Farrington Model currently implemented by the CDC for evaluating related disaster-attributable deaths [74–76], we employed an advanced flexible statistical framework that has higher power to detect the small and persistent increases in death rate introduced by such effects as one contiguous period [24]. Additionally, this modeling technique enabled us to account for the great across-TC variability and evaluated TC-specific POC and mortality risks, which tends to provide more precise effect estimates than the traditional multi-TC average estimates based on an identical exposure window. Finally, apart from providing the TCrelated mortality risk estimates like in most prior studies, we further estimated the POC, the ER curves between TC characteristics and mortality outcomes, as well as the temporal trends of mortality risks, which were important aspects for developing or adjusting disaster management policies and public health interventions to mitigate the adverse impacts of TCs.

There are also some limitations in our study. The possibility of residual confounding cannot be completely excluded. Despite our application of an advanced methodology to first attempt to estimate the TC-specific POC across countries, some uncertainty in the estimates caused by the limitations of this methodology must be acknowledged. We involved a predetermined wash-out period by excluding a certain length of post-TC period for other TCs (if any) when estimating the individual mortality risks for each TC event. The impacts of other TCs may still exist and bias our results. However, we believed the impact of this issue could be modest as evidenced by the robust results in sensitivity analyses by excluding different lengths of post-TC period. Furthermore, the determination of POC relied on statistical significance (i.e., the lower limit of the 95% CI > 0), potentially influenced by factors irrelevant to TCs, such as sample size. However, a series of simulation studies indicated that the impact of this issue on results may be modest, with consistent and reliable estimates in different scenarios [24]. Nonetheless, more studies are needed in the future to work on this challenging topic and improve the estimates of POC across TCs. In addition, mandatory pre-TC evacuation orders and population displacement during TC could be important influential factors in the health impacts of TCs. However, to our knowledge such data have never been systematically compiled and available on a multi-TC scale [24]. To minimize the health threats of climate and weather-related disasters, it is highly warranted to collect, compile, and incorporate richer data on these events in future studies. In this study, we estimated the yearly population size for each location and included it as an offset in the model to account for the long-term variations over time and across space due to the unavailability of daily population size in each location. Population characteristics such as the age and sex structure were associated with the vulnerability to TC hazards. Based on a classical two-stage modeling approach, we accounted for the temporal variations within a location by the control of temporal trends in the first stage, and the spatial variations across locations by using a mixed meta-regression model in the second stage. However, due to the lack of these data (e.g., age- or sex-specific daily mortality) at daily level and on a multicountry scale, we were thus unable to assess the potential risk differences across subpopulations such as different age and sex subgroups. Moreover, the TC-specific excess deaths for some TCs could be underestimated for some countries or territories without nationwide data such as China Mainland, Vietnam, and Mexico. The limited locations in China Mainland, Vietnam, and Mexico also increased the uncertainty of our results and prevented us from detecting significant findings for those countries and regions. We were also not able to estimate the TC-related mortality risk in the countries or territories without study locations but with a potentially high TC-related health burden (e.g., Bangladesh, Myanmar). These issues warrant further exploration with more comprehensive data and should be lessened in the future as the MCC network expands. Finally, land conditions could significantly influence the speed and direction of surface winds. The widely used improved wind field model by Holland incorporated an attenuation factor, the ratio between the distance to the center and the radius of maximum winds, to resemble surface friction effects [8,35-37]. This does not explicitly account for the surface friction-induced wind speed reduction [77] or motion-induced asymmetry [78]. Post-landfall, TC wind fields could become very noisy due to interaction with complex land surfaces, posing challenges in accounting for uncertainties when assessing the ER relationship with health outcomes [79]. However, neglecting inhomogeneous wind conditions over land is expected to minimally impact the results, given the study's focus on a binary TC exposure variable (exposed versus unexposed).

## Conclusion

To conclude, TCs show great variation in the POC and elevated mortality risks globally. The overall ER relationships of TC-related windspeed and rainfall with all-cause and cardiopulmonary mortality exhibited a monotonically increasing, non-threshold and linear curve, with a

heterogeneous pattern across regions. An overall decreasing trend was observed for the TCrelated all-cause and cardiovascular mortality risk from 1980 to 2019. The TC-related mortality risks were generally decreasing in most of the study countries, especially for the Philippines and the USA, while potentially increasing trends in TC-related all-cause and cardiovascular mortality risks were observed for Japan. Further targeted actions and in-depth explorations of TC epidemiology in the countries with high and increasing TC-related mortality burdens are particularly needed.

#### Supporting information

**S1 Text. MCC Collaborators.** (DOCX)

**S2 Text. Supplementary Material.** (DOCX)

#### Acknowledgments

We thank relevant institutes/agencies who provided data on mortality and weather conditions.

## **Author Contributions**

Conceptualization: Wenzhong Huang, Shanshan Li, Yuming Guo.

Data curation: Wenzhong Huang.

Formal analysis: Wenzhong Huang.

Funding acquisition: Yuming Guo.

Investigation: Wenzhong Huang.

Methodology: Wenzhong Huang, Thomas Vogt, Elizabeth A. Ritchie.

Resources: Yuming Guo.

Software: Wenzhong Huang.

Supervision: Yuming Guo.

Writing - original draft: Wenzhong Huang.

Writing – review & editing: Zhengyu Yang, Yiwen Zhang, Ben Armstrong, Wenhua Yu, Rongbin Xu, Pei Yu, Yanming Liu, Antonio Gasparrini, Samuel Hundessa, Eric Lavigne, Tomas Molina, Tobias Geiger, Yue Leon Guo, Christian Otto, Simon Hales, Farnaz Pourzand, Shih-Chun Pan, Ke Ju, Elizabeth A. Ritchie, Shanshan Li, Yuming Guo.

#### References

- Hudson P, Botzen WJW, Poussin J, Aerts JCJH. Impacts of Flooding and Flood Preparedness on Subjective Well-Being: A Monetisation of the Tangible and Intangible Impacts. J Happiness Stud. 2019; 20 (2):665–82.
- Parks RM, Guinto RR. Invited Perspective: Uncovering the Hidden Burden of Tropical Cyclones on Public Health Locally and Worldwide. Environ Health Perspect. 2022; 130(11):111306. https://doi.org/10. 1289/EHP12241 PMID: 36448793.
- 3. Geiger T, Frieler K, Bresch DN. A global historical data set of tropical cyclone exposure (TCE-DAT). Earth Syst Sci Data. 2018; 10(1):185–94.
- Chari F, Ngcamu BS, Novukela C. Supply chain risks in humanitarian relief operations: a case of Cyclone Idai relief efforts in Zimbabwe. J Humanit Logist Supply Chain Manag. 2021; 11(1):29–45.

- Ishizawa OA, Miranda JJ, Strobl E. The Impact of Hurricane Strikes on Short-Term Local Economic Activity: Evidence from Nightlight Images in the Dominican Republic. Int J Disaster Risk Sci. 2019; 10 (3):362–70.
- Wang S, Toumi R. Recent migration of tropical cyclones toward coasts. Science. 2021; 371(6528):514– 7. https://doi.org/10.1126/science.abb9038 PMID: 33510027.
- Knutson T, Camargo SJ, Chan JCL, Emanuel K, Ho C-H, Kossin J, et al. Tropical Cyclones and Climate Change Assessment: Part II: Projected Response to Anthropogenic Warming. Bull Am Meteorol Soc. 2020; 101(3):E303–E22.
- Geiger T, Gütschow J, Bresch DN, Emanuel K, Frieler K. Double benefit of limiting global warming for tropical cyclone exposure. Nat Clim Change. 2021; 11(10):861–6.
- Parks RM, Benavides J, Anderson GB, Nethery RC, Navas-Acien A, Dominici F, et al. Association of Tropical Cyclones With County-Level Mortality in the US. JAMA. 2022; 327(10):946–55. <u>https://doi.org/ 10.1001/jama.2022.1682</u> PMID: 35258534.
- Parks RM, Anderson GB, Nethery RC, Navas-Acien A, Dominici F, Kioumourtzoglou MA. Tropical cyclone exposure is associated with increased hospitalization rates in older adults. Nat Commun. 2021; 12(1):1545. https://doi.org/10.1038/s41467-021-21777-1 PMID: 33750775.
- Yan M, Wilson A, Dominici F, Wang Y, Al-Hamdan M, Crosson W, et al. Tropical Cyclone Exposures and Risks of Emergency Medicare Hospital Admission for Cardiorespiratory Diseases in 175 Urban United States Counties, 1999–2010. Epidemiology. 2021; 32(3):315–26. <u>https://doi.org/10.1097/EDE.</u> 000000000001337 PMID: 33591048.
- Sharpe JD, Wolkin AF. The Epidemiology and Geographic Patterns of Natural Disaster and Extreme Weather Mortality by Race and Ethnicity, United States, 1999–2018. Public Health Rep. 2022; 137 (6):1118–25. https://doi.org/10.1177/00333549211047235 PMID: 34678107.
- Dresser C, Hart A, Kwok-Keung Law A, Yen Yen Poon G, Ciottone G, Balsari S. Where are People Dying in Disasters, and Where is it Being Studied? A Mapping Review of Scientific Articles on Tropical Cyclone Mortality in English and Chinese. Prehosp Disaster Med. 2022; 37(3):409–16. https://doi.org/ 10.1017/S1049023X22000541 PMID: 35379375.
- Hu P, Zhang Q, Shi P, Chen B, Fang J. Flood-induced mortality across the globe: Spatiotemporal pattern and influencing factors. Sci Total Environ. 2018; 643:171–82. <u>https://doi.org/10.1016/j.scitotenv.</u> 2018.06.197 PMID: 29936160.
- Begum TF, Lin Z, Primeau M, Lin S. Assessing short-term and long-term mental health effects among older adults after Hurricane Sandy. Sci Total Environ. 2022; 825:153753. <u>https://doi.org/10.1016/j. scitotenv.2022.153753</u> PMID: 35151740.
- Bell SA, Donnelly JP, Li W, Davis MA. Hospitalizations for chronic conditions following hurricanes among older adults: A self-controlled case series analysis. J Am Geriatr Soc. 2022; 70(6):1695–703. https://doi.org/10.1111/jgs.17702 PMID: 35171505.
- Huang W, Gao Y, Xu R, Yang Z, Yu P, Ye T, et al. Health Effects of Cyclones: A Systematic Review and Meta-Analysis of Epidemiological Studies. Environ Health Perspect. 2023; 131(8):86001. <u>https://doi.org/10.1289/EHP12158 PMID</u>: 37639476.
- Cruz-Cano R, Mead EL. Causes of excess deaths in Puerto Rico after Hurricane Maria: a time-series estimation. Am J Public Health. 2019; 109(7):1050–2. <u>https://doi.org/10.2105/AJPH.2019.305015</u> PMID: 30998411.
- Santos-Burgoa C, Sandberg J, Suarez E, Goldman-Hawes A, Zeger S, Garcia-Meza A, et al. Differential and persistent risk of excess mortality from Hurricane Maria in Puerto Rico: a time-series analysis. Lancet Planet Health. 2018; 2(11):e478–e88. <u>https://doi.org/10.1016/S2542-5196(18)30209-2</u> PMID: 30318387.
- Santos-Lozada AR, Howard JT. Use of death counts from vital statistics to calculate excess deaths in Puerto Rico following Hurricane Maria. JAMA. 2018; 320(14):1491–3. https://doi.org/10.1001/jama. 2018.10929 PMID: 30073274.
- 21. Rivera R, Rolke W. Estimating the death toll of Hurricane Maria. Wiley Online Library; 2018.
- Kishore N, Marques D, Mahmud A, Kiang MV, Rodriguez I, Fuller A, et al. Mortality in puerto rico after hurricane maria. N Engl J Med. 2018; 379(2):162–70. https://doi.org/10.1056/NEJMsa1803972 PMID: 29809109.
- 23. Marazzi M, Miloucheva B, Bobonis GJ. Mortality of Puerto Ricans in the USA post Hurricane Maria: an interrupted time series analysis. BMJ Open. 2022; 12(8):e058315. <u>https://doi.org/10.1136/bmjopen-2021-058315</u> PMID: 36041757.
- Acosta RJ, Irizarry RA. A Flexible Statistical Framework for Estimating Excess Mortality. Epidemiology. 2022; 33(3):346–53. https://doi.org/10.1097/EDE.00000000001445 PMID: 35383642.

- Rivera-Hernandez M, Kim D, Nguyen KH, Thorsness R, Lee Y, Swaminathan S, et al. Changes in Migration and Mortality Among Patients With Kidney Failure in Puerto Rico After Hurricane Maria. JAMA Health Forum. 2022; 3(8):e222534. https://doi.org/10.1001/jamahealthforum.2022.2534 PMID: 36200633.
- Sun S, Weinberger KR, Yan M, Brooke Anderson G, Wellenius GA. Tropical cyclones and risk of preterm birth: A retrospective analysis of 20 million births across 378 US counties. Environ Int. 2020; 140:105825. https://doi.org/10.1016/j.envint.2020.105825 PMID: 32485474.
- Nethery RC, Katz-Christy N, Kioumourtzoglou MA, Parks RM, Schumacher A, Anderson GB. Integrated causal-predictive machine learning models for tropical cyclone epidemiology. Biostatistics. 2023; 24 (2):449–64. https://doi.org/10.1093/biostatistics/kxab047 PMID: 34962265.
- Gasparrini A, Guo Y, Hashizume M, Lavigne E, Zanobetti A, Schwartz J, et al. Mortality risk attributable to high and low ambient temperature: a multicountry observational study. Lancet. 2015; 386 (9991):369–75. https://doi.org/10.1016/S0140-6736(14)62114-0 PMID: 26003380.
- Zhao Q, Guo Y, Ye T, Gasparrini A, Tong S, Overcenco A, et al. Global, regional, and national burden of mortality associated with non-optimal ambient temperatures from 2000 to 2019: a three-stage modelling study. Lancet Planet Health. 2021; 5(7):e415–e25. https://doi.org/10.1016/S2542-5196(21)00081-4 PMID: 34245712.
- Nguyen H, Yee KC, Braude M, Moldovan C, Cocker F, Palmer AJ, et al. Accuracy of coded cause of death data: a study based on primary liver cancer. Tasman Med J. 2022; 4(2):12–20.
- Mortality Collection data dictionary: New Zealand Ministry of Health. 2021. Available from: https://www. health.govt.nz/publication/mortality-collection-data-dictionary.
- Morais RMd, Costa AL. Uma avaliação do Sistema de Informações sobre Mortalidade. Saúde em Debate. 2017; 41(spe):101–17.
- Hebbern C, Gosselin P, Chen K, Chen H, Cakmak S, MacDonald M, et al. Future temperature-related excess mortality under climate change and population aging scenarios in Canada. Can J Public Health. 2023; 114(5):726–36. https://doi.org/10.17269/s41997-023-00782-5 PMID: 37308698.
- Murakami D, Yamagata Y. Estimation of Gridded Population and GDP Scenarios with Spatially Explicit Statistical Downscaling. Sustainability. 2019; 11(7):2106.
- 35. Holland G. A Revised Hurricane Pressure–Wind Model. Mon Weather Rev. 2008; 136(9):3432–45.
- Peduzzi P, Chatenoux B, Dao H, De Bono A, Herold C, Kossin J, et al. Global trends in tropical cyclone risk. Nat Clim Change. 2012; 2(4):289–94.
- Lange S, Volkholz J, Geiger T, Zhao F, Vega I, Veldkamp T, et al. Projecting Exposure to Extreme Climate Impact Events Across Six Event Categories and Three Spatial Scales. Earth's Future. 2020; 8 (12):e2020EF001616.
- Knapp KR, Kruk MC, Levinson DH, Diamond HJ, Neumann CJ. The International Best Track Archive for Climate Stewardship (IBTrACS): Unifying Tropical Cyclone Data. Bull Am Meteorol Soc. 2010; 91 (3):363–76.
- Aznar-Siguan G, Bresch DN. CLIMADA v1: a global weather and climate risk assessment platform. Geosci Model Dev. 2019; 12(7):3085–97.
- Hersbach H, Bell B, Berrisford P, Biavati G, Horányi A, Muñoz Sabater J, et al. ERA5 hourly data on single levels from 1979 to present. Copernicus climate change service (c3s) climate data store (cds). 2018; 10(10.24381).
- Anderson GB, Schumacher A, Done J. Exposure Assessment for Tropical Cyclone Epidemiology. Curr Environ Health Rep. 2022; 9(1):104–19. https://doi.org/10.1007/s40572-022-00333-z PMID: 35167050.
- Schenkel BA, Hart RE. An Examination of Tropical Cyclone Position, Intensity, and Intensity Life Cycle within Atmospheric Reanalysis Datasets. J Climate. 2012; 25(10):3453–75.
- Zick SE, Matyas CJ. Tropical cyclones in the North American Regional Reanalysis: An assessment of spatial biases in location, intensity, and structure. J Geophys Res Atmos. 2015; 120(5):1651–69.
- Saha S, Moorthi S, Wu X, Wang J, Nadiga S, Tripp P, et al. The NCEP Climate Forecast System Version 2. J Climate. 2014; 27(6):2185–208.
- **45.** Chua PLC, Ng CFS, Madaniyazi L, Seposo X, Salazar MA, Huber V, et al. Projecting Temperature-Attributable Mortality and Hospital Admissions due to Enteric Infections in the Philippines. Environ Health Perspect. 2022; 130(2):27011. https://doi.org/10.1289/EHP9324 PMID: 35188405.
- 46. Kiang MV, Carlasare LE, Thadaney Israni S, Norcini JJ, Zaman JAB, Bibbins-Domingo K. Excess Mortality Among US Physicians During the COVID-19 Pandemic. JAMA Intern Med. 2023; 183(4):374–6. https://doi.org/10.1001/jamainternmed.2022.6308 PMID: 36745424.

- Kiang MV, Acosta RJ, Chen YH, Matthay EC, Tsai AC, Basu S, et al. Sociodemographic and geographic disparities in excess fatal drug overdoses during the COVID-19 pandemic in California: A population-based study. Lancet Reg Health Am. 2022; 11:100237. https://doi.org/10.1016/j.lana.2022. 100237 PMID: 35342895.
- 48. R Core Team R. R: A language and environment for statistical computing. 2013.
- Irizarry RA, Acosta R. excessmort: Excess Mortality. R package version 0.6.1. 2021. Available from: https://CRAN.R-project.org/package=excessmort.
- Sera F, Armstrong B, Blangiardo M, Gasparrini A. An extended mixed-effects framework for meta-analysis. Stat Med. 2019; 38(29):5429–44. https://doi.org/10.1002/sim.8362 PMID: 31647135.
- Li C, Zhao Z, Yan Y, Liu Q, Zhao Q, Ma W. Short-term effects of tropical cyclones on the incidence of dengue: a time-series study in Guangzhou, China. Parasit Vectors. 2022; 15(1):358. https://doi.org/10. 1186/s13071-022-05486-2 PMID: 36203178.
- Utsumi N, Kim H. Observed influence of anthropogenic climate change on tropical cyclone heavy rainfall. Nat Clim Change. 2022; 12(5):436–40.
- 53. Robinson Md MPHTD, Oliveira Md MPHTM, Kayden Md MPHS. Factors affecting the United Nations' response to natural disasters: what determines the allocation of the Central Emergency Response Fund? Disasters. 2017; 41(4):631–48. https://doi.org/10.1111/disa.12226 PMID: 28133779
- 54. Kusonwattana P, Ong AKS, Prasetyo YT, Mariñas KA, Yuduang N, Chuenyindee T, et al. Predicting Factors Affecting the Intention to Prepare for Mitigation of Man-Made Fire Disasters in Chonburi Province, Thailand: An Integration of Structural Equation Modeling and Artificial Neural Network Hybrid Approach. Sustainability. 2022; 14(22):15442.
- 55. Jeong S, Yoon DK. Examining Vulnerability Factors to Natural Disasters with a Spatial Autoregressive Model: The Case of South Korea. Sustainability. 2018; 10(5):1651.
- Garner AJ, Kopp RE, Horton BP. Evolving Tropical Cyclone Tracks in the North Atlantic in a Warming Climate. Earth's Future. 2021; 9(12):e2021EF002326.
- Kossin JP, Olander TL, Knapp KR. Trend Analysis with a New Global Record of Tropical Cyclone Intensity. J Climate. 2013; 26(24):9960–76.
- Altman J, Ukhvatkina ON, Omelko AM, Macek M, Plener T, Pejcha V, et al. Poleward migration of the destructive effects of tropical cyclones during the 20th century. Proc Natl Acad Sci U S A. 2018; 115 (45):11543–8. https://doi.org/10.1073/pnas.1808979115 PMID: 30348774.
- Lin TC, Hogan JA, Chang CT. Tropical Cyclone Ecology: A Scale-Link Perspective. Trends Ecol Evol. 2020; 35(7):594–604. https://doi.org/10.1016/j.tree.2020.02.012 PMID: 32521243.
- Chen C, Noble I, Hellmann J, Coffee J, Murillo M, Chawla N. University of Notre Dame global adaptation index. Notre Dame, IN, USA: University of Notre Dame, 2015.
- 61. Moon IJ, Kim SH, Chan JCL. Climate change and tropical cyclone trend. Nature. 2019; 570(7759):E3– e5. https://doi.org/10.1038/s41586-019-1222-3 PMID: 31168110.
- Yamaguchi M, Chan JCL, Moon IJ, Yoshida K, Mizuta R. Global warming changes tropical cyclone translation speed. Nat Commun. 2020; 11(1):47. https://doi.org/10.1038/s41467-019-13902-y PMID: 31913276.
- Park D-SR, Ho C-H, Nam CC, Kim H-S. Evidence of reduced vulnerability to tropical cyclones in the Republic of Korea. Environ Res Lett. 2015; 10(5):054003.
- Wang S, Toumi R. More tropical cyclones are striking coasts with major intensities at landfall. Sci Rep. 2022; 12(1):5236. https://doi.org/10.1038/s41598-022-09287-6 PMID: 35347203.
- **65.** Wang S, Toumi R. On the intensity decay of tropical cyclones before landfall. Sci Rep. 2022; 12 (1):3288. https://doi.org/10.1038/s41598-022-07310-4 PMID: 35228600.
- 66. Šakić Trogrlić R, van den Homberg M, Budimir M, McQuistan C, Sneddon A, Golding B. Early Warning Systems and Their Role in Disaster Risk Reduction. In: Golding B, editor. Towards the "Perfect" Weather Warning: Bridging Disciplinary Gaps through Partnership and Communication. Cham: Springer International Publishing; 2022. p. 11–46.
- Mashi SA, Oghenejabor OD, Inkani AI. Disaster risks and management policies and practices in Nigeria: A critical appraisal of the National Emergency Management Agency Act. Int J Disaster Risk Reduct. 2019; 33:253–65.
- Keim ME. Building human resilience: the role of public health preparedness and response as an adaptation to climate change. Am J Prev Med. 2008; 35(5):508–16. <u>https://doi.org/10.1016/j.amepre.2008.08.</u> 022 PMID: 18929977.
- Mensah GA, Roth GA, Fuster V. The Global Burden of Cardiovascular Diseases and Risk Factors: 2020 and Beyond. J Am Coll Cardiol. 2019; 74(20):2529–32. https://doi.org/10.1016/j.jacc.2019.10.009 PMID: 31727292.

- 70. Cheng X, Yang Y, Schwebel DC, Liu Z, Li L, Cheng P, et al. Population ageing and mortality during 1990–2017: A global decomposition analysis. PLoS Med. 2020; 17(6):e1003138. https://doi.org/10. 1371/journal.pmed.1003138 PMID: 32511229.
- Studholme J, Fedorov AV, Gulev SK, Emanuel K, Hodges K. Poleward expansion of tropical cyclone latitudes in warming climates. Nat Geosci. 2022; 15(1):14–28.
- 72. Bi K, Xie L, Zhang H, Chen X, Gu X, Tian Q. Accurate medium-range global weather forecasting with 3D neural networks. Nature. 2023; 619(7970):533–8. https://doi.org/10.1038/s41586-023-06185-3 PMID: 37407823.
- Kossin JP, Knapp KR, Olander TL, Velden CS. Global increase in major tropical cyclone exceedance probability over the past four decades. Proc Natl Acad Sci U S A. 2020; 117(22):11975–80. https://doi. org/10.1073/pnas.1920849117 PMID: 32424081.
- 74. Center for Disease and Prevention. Excess Deaths Associated with COVID-19. 2020. Available from: https://www.cdc.gov/nchs/nvss/vsrr/covid19/excess\_deaths.htm#techNotes.
- Noufaily A, Enki DG, Farrington P, Garthwaite P, Andrews N, Charlett A. An improved algorithm for outbreak detection in multiple surveillance systems. Stat Med. 2013; 32(7):1206–22. https://doi.org/10.1002/sim.5595 PMID: 22941770.
- Farrington CP, Andrews NJ, Beale AD, Catchpole MA. A Statistical Algorithm for the Early Detection of Outbreaks of Infectious Disease. J R Stat Soc A Stat Soc. 1996; 159(3):547–63.
- 77. Lin N, Chavas D. On hurricane parametric wind and applications in storm surge modeling. J Geophys Res Atmos. 2012; 117:D09120.
- Uhlhorn EW, Klotz BW, Vukicevic T, Reasor PD, Rogers RF. Observed Hurricane Wind Speed Asymmetries and Relationships to Motion and Environmental Shear. Mon Weather Rev. 2014; 142(3):1290–311.
- **79.** Knaff JA, Sampson CR, Kucas ME, Slocum CJ, Brennan MJ, Meissner T, et al. Estimating tropical cyclone surface winds: Current status, emerging technologies, historical evolution, and a look to the future. Trop Cyclone Res Rev. 2021; 10(3):125–50.